SEKS-SSR-EV- /Week 5 starting   29 Jan 07
SM18 Operation Meeting (Tuesdays) 13 Feb  07

Bench status  on Monday 12-Feb-2007 
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Situation on the test benches ;  

-MB2267, MB2425 warmed up. 2 MB for the machines:
-MB2426 (on test) and MB2394 (coming on E1 bench).

-2 500- SSS to be tested : SSS529 (thermal cycle) and SSS530 cooling down

-1 Arc SSS on cold test (SSS374)

-MM on SSS374 and SSS529 (restricted MM program).

Situation from Cryogenics:  J-P Lamboy (connector damaged on D1, long cooling in TBF2)  
Situation of mechanics  : J Halik 

    General Points:

· Test coordination by M. Buzio from Wednesday mid-day.
· MQTL in SSS531: Stopped at 9 training quenches,500 A (not nominal, 300 A enough)
· Thermal cycle rules for SSS529 (not conform for MQ and MQTL).
4  Magnet Tests completed () + 1 run on SSS529; 

2 Dipoles tests


2 arc SSS 
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Quench Information
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 Case of SSS529 (MQ)
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Wednesday_07_M
	TBF2
	MBAR 2267 was not cooling down, Cryo replied that magnet was in the same phase as in night after quench and they will started cooling now.
	ACR
	
	Tuning of the cryo parameters to be done. One night lost.

	Wednesday_07_M
	TBB2
	SSS531: CD-corrector powering could not be done. PC Aux error in PLCSE.
	MTM
	
	Solved by Salem.

	Wednesday_07_M
	TBE2
	MB2426 :Leak found during capillary test (M2 line)
	ACR
	
	Joint with default. Lionel checked the other ones available for ICS. Magnet disconnected and reconnected.

	Wednesday_07_A
	TBB1
	SSS374: Abnormal voltage tap signal shapes during IAP (Prep 2.2). 
	MTM
	
	Salem contacted. He informed that the configuration files need to be corrected for this magnet. Done in the next morning.

	Wednesday_07_A
	TBB1
	SSS374: HV could not be launched due to wrong cryo phase interlock.
	ACR
	
	Salem solved the problem with ACR team next morning.

	Wednesday_07_A
	TBD1
	SSS529: EF corrector powering – 600A power converter is not going to standby mode, shows some 'bad state error'. 
	AB/PO
	
	Salem contacted. He reset the power supply.

	Thursday_08_M
	TBB1
	SSS374 : CQW1 HV cannot be uploaded. Link to MTF from this is not available, may be blocked
	TS
	
	

	Thursday_08_A
	
	In HV rack MY95 temperature sensor and humidity sensor not found.
	MTM
	
	

	Friday_09_M
	TBF2
	MB2267: Loss of cryo-OK during LHC cycle.
	ACR
	
	Return BP to high. Quench and test to be re done.

	Friday_09_M
	TBA1
	MB1334: Lyra call came without WP04.
	MTM
	
	Stéphane asked to go on with connection WP04 was executed by Richard.

	Friday_09_A
	TBD1
	SSS529: Connectors on the CFB damaged.
	TCR
	
	SSS warming up naturally. Thermal cycle executed Monday after repair.

	Monday_12_A
	TBF1
	MB2425: HV test AP, HV Rack (MY92) unable to launch HV Test (Dipole-Ground) due to PLC-HV communication failure.
	
	
	Solved by Salem, changing the HV rack.

































































